Author’s response to reviews

Title: The Child Health Utility 9D Index as an outcome measure for child dental health.

Authors:

Lyndie A Foster Page (lyndie.fosterpage@otago.ac.nz)
W M Thomson (murray.thomson@otago.ac.nz)
Zoe Marshman (Z.Marshman@sheffield.ac.uk)
Katherine J Stevens (k.stevens@sheffield.ac.uk)

Version: 2 Date: 29 April 2014

Author’s response to reviews: see over
The Editor
BMC Oral Health

Dear Dr Morawska

We re-submit the manuscript of our article “The Child Health Utility 9D Index as an outcome measure for child dental health.” for consideration for publication in BMC Oral Health. We have now formatted the paper to the guidelines required by your journal.

We consider that it is appropriate for BMC Oral Health as the findings are relevant for an international audience. We have no prior or duplicate publication or submission elsewhere of any part of the work. We have no statement of financial relationships that might lead to a conflict of interest and the manuscript has been read and approved by the other corresponding authors. I declare that I am a deputy section editor for BMC Oral Health and understand this may cause a conflict of interest. We have no issues regarding your editorial policies. We believe that the manuscript represents honest work conducted in New Zealand.

We look forward to hearing from you.

Yours sincerely

Authors

Dr L A Foster Page, Professor WM Thomson, Dr Z Marshman, Dr K Stevens

Corresponding author
L A Foster Page  (064 34795853)
Department of Oral Rehabilitation
The University of Otago
PO Box 647
Dunedin, New Zealand lyndie.fosterpage@otago.ac.nz